Author's response to reviews

Title: Efficacy of strength and aerobic exercise on patient-reported outcomes and structural changes in patients with knee osteoarthritis. Study protocol for a randomized controlled trial

Authors:

Britt Elin Øiestad (britt.e.oiestad@nimi.no)
Nina Østerås (nina.osteras@medisin.uio.no)
Richard Frobell (richard.frobell@med.lu.se)
Margreth Grotle (margreth.grotle@medisin.uio.no)
Helga Brøgger (helga@brogger.no)
May Arna Risberg (m.a.risberg@nih.no)

Version: 2 Date: 12 August 2013

Author's response to reviews:

Dear reviewers,

I have now included a "Competing interests" section, and a Conclusion section. The changes are marked in the manuscript with a date of the changes.

Please let me know if this was not what you intended.

Sincerely,

Britt Elin Øiestad